Author's response to reviews

Title: The Use of Complementary and Alternative Medicine (CAM) by African American (AA) and Caucasian American (CA) Elders in a Rural Setting: a Descriptive, Comparative Study

Authors:

Norma G. Cuellar R.N., D.S.N., CCRN (mattdarat7@aol.com)
Teresa Aycock R.N., M.S.N. (tomon99@aol.com)
Bridgett Cahill R.N., M.S.N., FNP (mbcahill1@megagate.com)
Julie Ford R.N., M.S.N., FNP (jfx2@megagate.com)

Version: 4 Date: 17 Nov 2003

Thank you for accepting the article. I have made all the revisions that were suggested, including the few from the reviewers and the requests for editing purposes. I have made no changes to the reference page since I have used Endnote as the author guidelines recommend. I hope this is OK.

Thank you again.

Dr. Cuellar